

International Journal of Innovative Technologies in Social Science

e-ISSN: 2544-9435

Scholarly Publisher RS Global Sp. z O.O. ISNI: 0000 0004 8495 2390

Dolna 17, Warsaw, Poland 00-773 +48 226 0 227 03 editorial office@rsglobal.pl

ARTICLE TITLE	REPAIRING GENES, RESTORING STRENGTH: THE PROMISE OF MODERN GENE THERAPY IN DUCHENNE MUSCULAR DYSTROPHY				
ARTICLE INFO	Katarzyna Kozon, Agnieszka Floriańczyk, Ewa Romanowicz, Aleksandra Kołdyj, Agnieszka Ozdarska, Adrian Krzysztof Biernat, Marcin Lampart, Anna Rupińska, Zuzanna Jasińska, Kamila Krzewska. (2025) Repairing Genes, Restoring Strength: The Promise of Modern Gene Therapy in Duchenne Muscular Dystrophy. <i>International Journal of Innovative Technologies in Social Science</i> . 3(47). doi: 10.31435/ijitss.3(47).2025.3468				
DOI	https://doi.org/10.31435/ijitss.3(47).2025.3468				
RECEIVED	08 June 2025				
ACCEPTED	14 July 2025				
PUBLISHED	16 July 2025				
LICENSE	The article is licensed under a Creative Commons Attribution 4.0 International License.				

© The author(s) 2025.

This article is published as open access under the Creative Commons Attribution 4.0 International License (CC BY 4.0), allowing the author to retain copyright. The CC BY 4.0 License permits the content to be copied, adapted, displayed, distributed, republished, or reused for any purpose, including adaptation and commercial use, as long as proper attribution is provided.

REPAIRING GENES, RESTORING STRENGTH: THE PROMISE OF MODERN GENE THERAPY IN DUCHENNE MUSCULAR DYSTROPHY

Katarzyna Kozon (Corresponding Author, Email: kasia.kozon1066@gmail.com) Masovian Bródnowski Hospital, ul. Ludwika Kondratowicza 8, 03-242 Warsaw, Poland ORCID ID: 0009-0009-6536-946X

Agnieszka Floriańczyk

Masovian Bródnowski Hospital, ul. Ludwika Kondratowicza 8, 03-242 Warsaw, Poland ORCID ID: 0009-0003-5136-0380

Ewa Romanowicz

Independent Public Healthcare Institution in Wyszków, ul. Komisji Edukacji Narodowej 1, 07–200 Wyszków, Poland ORCID ID: 0009-0004-6612-3174

Aleksandra Kołdyj

District Medical Centre in Grójec, ul. Piotra Skargi 10, 05-600 Grójec, Poland ORCID ID: 0009-0002-5695-608X

Agnieszka Ozdarska

Independent Public Healthcare Institution in Wyszków, ul. Komisji Edukacji Narodowej 1, 07–200 Wyszków, Poland ORCID ID: 0009-0008-7083-2140

Adrian Krzysztof Biernat

Specialist Provincial Hospital in Ciechanów, ul. Powstańców Wielkopolskich 2, 06-400 Ciechanów, Poland ORCID ID: 0009-0007-6734-0447

Marcin Lampart

St. John Paul II Independent Public Western Specialist Hospital, ul. Daleka 11, 05-825 Grodzisk Mazowiecki, Poland ORCID ID: 0009-0005-8485-850X

Anna Rupińska

Independent Public Outpatient Healthcare Group – Warsaw-Ochota, ul. Szczęśliwicka 36, 02-353 Warsaw, Poland ORCID ID: 0009-0001-8567-5925

Zuzanna Jasińska

Warsaw Southern Hospital, ul. Rotmistrza Witolda Pileckiego 99, 02-781 Warsaw, Poland ORCID ID: 0009-0001-9448-782X

Kamila Krzewska

Primary Care Clinic "Zdrowa Rodzina", ul. Generała Tadeusza Pełczyńskiego 22J, 01-471 Warsaw, Poland ORCID ID: 0009-0002-5672-3796

ABSTRACT

Introduction: Duchenne muscular dystrophy (DMD) is a severe X-linked neuromuscular disorder characterized by progressive muscle degeneration due to mutations in the dystrophin gene. Current treatments primarily focus on symptom management. Delandistrogene moxeparvovec is a gene therapy designed to deliver a shortened, functional version of the dystrophin gene, potentially modifying disease progression.

Aim of the study: To evaluate the clinical efficacy and safety of delandistrogene moxeparvovec in improving skeletal muscle function and physical performance in patients with DMD based on data from completed clinical trials.

Methodology: The analysis includes four published clinical trials involving ambulatory boys aged 4–8 years with confirmed dystrophin mutations. Primary outcomes focused on changes in the North Star Ambulatory Assessment (NSAA), while secondary endpoints included time to rise from the floor and the 10-meter walk/run test. Study designs ranged from open-label to randomized, double-blind trials. Adverse events and regulatory outcomes were also considered.

Conclusions: Delandistrogene moxeparvovec demonstrated variable improvements in motor function, with more pronounced benefits in younger age groups. Some trials showed statistically significant results, while others failed to meet primary endpoints. Adverse events, including serious complications, were reported. The therapy received FDA approval in 2023 and 2024 for defined patient subgroups, though some decisions were made despite inconclusive efficacy data.

KEYWORDS

Duchenne Muscular Dystrophy, Delandistrogene Moxeparvovec, Gene Therapy, Skeletal Muscle Function, Genetic Disorders, Dystrophin

CITATION

Katarzyna Kozon, Agnieszka Floriańczyk, Ewa Romanowicz, Aleksandra Kołdyj, Agnieszka Ozdarska, Adrian Krzysztof Biernat, Marcin Lampart, Anna Rupińska, Zuzanna Jasińska, Kamila Krzewska. (2025) Repairing Genes, Restoring Strength: The Promise of Modern Gene Therapy in Duchenne Muscular Dystrophy. *International Journal of Innovative Technologies in Social Science*. 3(47). doi: 10.31435/ijitss.3(47).2025.3468

COPYRIGHT

© The author(s) 2025. This article is published as open access under the Creative Commons Attribution 4.0 International License (CC BY 4.0), allowing the author to retain copyright. The CC BY 4.0 License permits the content to be copied, adapted, displayed, distributed, republished, or reused for any purpose, including adaptation and commercial use, as long as proper attribution is provided.

1. Introduction

Duchenne muscular dystrophy (DMD) is the most common and one of the most severe forms of muscular dystrophy, inherited in an X-linked recessive manner (Bez Batti Angulski et al., 2023). The disease results from mutations in the dystrophin (DMD) gene, leading to progressive damage of striated skeletal and cardiac muscle tissue (Shih et al., 2020). Dystrophin is a key component of the sarcolemmal protein complex, which stabilizes and protects muscle fibers during contraction. It functions as a structural bridge between the myofibrillar cytoskeleton and the extracellular matrix. The absence or dysfunction of dystrophin gradually compromises sarcolemmal integrity, resulting in progressive muscle fiber degeneration and pathological fat accumulation (Aldharee, 2024). Clinically, DMD is characterized by early symptom onset, typically around 2-3 years of age. By the age of 12, most patients become wheelchair-dependent, and by approximately 20 years of age, assisted ventilation becomes necessary. The average life expectancy ranges from 20 to 40 years, with respiratory or cardiac failure being the most common causes of death (Duan et al., 2021). The diagnostic process for DMD usually begins in early childhood, prompted by hallmark clinical signs such as a positive Gowers' sign, progressive muscle weakness, and motor clumsiness. Muscle atrophy typically begins in the proximal limb muscles and progresses to distal groups. A distinctive feature of DMD is calf pseudohypertrophy caused by fibrotic tissue deposition and fat infiltration. As the disease progresses, children often develop toewalking due to shortening of the plantar flexor muscles. Less frequently, diagnosis is preceded by the detection of developmental delay or elevated serum enzyme levels, such as alanine aminotransferase (ALT), aspartate aminotransferase (AST), lactate dehydrogenase (LDH), or creatine kinase (CK). Elevated ALT, AST, or LDH may be misattributed to hepatic pathology, potentially delaying the correct diagnosis of DMD (Birnkrant et al., 2018; Patterson et al., 2023). Conventional DMD therapies have focused on symptom management and delaying disease progression through glucocorticoids, physical rehabilitation, and support of respiratory and cardiovascular function (Trucco et al., 2020). However, these treatments do not target the underlying genetic defect and offer only partial clinical benefit. In recent years, gene therapy has emerged as a promising treatment avenue, aiming to introduce functional copies of the dystrophin gene or its truncated variants such as microdystrophin—capable of partially restoring protein function (Mendell et al., 2023). Delandistrogene moxeparvovec gene therapy (Elevidys, SRP-9001) is a gene therapy currently undergoing clinical evaluation. Its mechanism involves delivering a gene encoding micro-dystrophin to the patient's muscle cells via a viral vector, enabling synthesis of a functional protein and potentially improving muscle structure and function (Mendell, Sahenk, et al., 2024).

2. Aim of The Study

This review aims to assess the therapeutic potential of delandistrogene moxeparvovec (Elevidys) gene therapy in the treatment of Duchenne muscular dystrophy (DMD). As previously outlined, DMD is a progressive genetic disorder that causes degeneration of skeletal, respiratory, and cardiac muscles, significantly reducing lifespan and quality of life. Despite the availability of symptomatic treatments, conventional therapies do not modify the molecular progression of the disease. Delandistrogene moxeparvovec, the first approved gene therapy for DMD, offers the possibility of delivering a functional version of the dystrophin gene, thereby opening new therapeutic prospects. This review presents the current understanding of the mechanism of action of this therapy, its impact on motor function and clinical outcomes, and discusses potential limitations and challenges associated with its application. The primary goal is to determine whether delandistrogene moxeparvovec represents a transformative therapeutic intervention that alters the natural course of DMD and improves patient quality of life.

3. Methodology

This review is based on an analysis of the available scientific literature concerning Elevidys gene therapy and its application in the treatment of Duchenne muscular dystrophy. Literature searches were primarily conducted using the PubMed database, employing a combination of the following keywords: "Duchenne muscular dystrophy", "gene therapy", "Elevidys", "delandistrogene moxeparvovec", "microdystrophin", "AAV gene delivery", "functional outcomes", "motor function", "quality of life", and "clinical trial." Studies published in peer-reviewed journals between 2010 and 2025 were included, with particular emphasis on clinical trials conducted between 2022 and 2025. Included in the analysis were phase 1–3 clinical trials, systematic reviews, and expert articles focusing on the mechanism of action, safety, and efficacy of Elevidys. Special attention was given to the therapy's effects on functional metrics (e.g., North Star Ambulatory Assessment), biomarkers of dystrophin expression, and quality of life indicators for patients and their families. Selection criteria for the literature included recency, scientific relevance, methodological clarity, and data reliability.

4. Current State of Knowledge

4.1 Molecular Mechanisms of DMD and the Role of Muscle Function in Daily Living

Duchenne muscular dystrophy is caused by mutations in the DMD gene, which, as previously discussed, result in the absence of the muscle-specific dystrophin isoform (Dp427m). Mutations in the DMD gene may also lead to Becker muscular dystrophy (BMD), a milder disorder characterized by later onset and slower progression (Mercuri et al., 2019). Thousands of DMD gene mutations have been identified in patients with Duchenne and Becker muscular dystrophies. In both disorders, deletions account for approximately 60–70% of mutations. In DMD, duplications represent 5-15% of cases, while point mutations, small deletions, or insertions make up about 20%; the distribution in BMD is slightly different. These mutations commonly cluster in so-called hotspot regions, including exons 45-55 and 3-9 for DMD. Rarely, larger genomic rearrangements such as translocations between the X chromosome and autosomes occur, causing DMD in both males and females (Nevin et al., 1986). The DMD gene encodes the full-length dystrophin isoform Dp427m found in muscle, as well as other full-length isoforms- Dp427c and Dp427p present in cortical neurons and Purkinje cells of the cerebellum, respectively. In addition to these, shorter dystrophin isoforms are expressed in various tissues (Chelly et al., 1990; Górecki et al., 1992). Dystrophin forms a complex with associated proteins collectively known as the dystrophin-associated protein complex (DAPC) which links the sarcolemma (muscle cell membrane) to the cytoskeleton and extracellular matrix (Gao & McNally, 2015). The absence of dystrophin leads to the disintegration of the DAPC, disrupting the connection between the cytoskeleton, sarcolemma, and extracellular matrix. The sarcolemma becomes susceptible to mechanical damage during muscle contractions, facilitating the leakage of muscle enzymes into the bloodstream, such as creatine kinase (Aartsma-Rus & van Putten, 2014). Dystrophin is also involved in nitric oxide production and vasodilation during exercise. In DMD, these processes are impaired,

leading to muscle ischemia during exertion (Sander et al., 2000). Moreover, muscles in DMD produce increased levels of reactive oxygen and nitrogen species, while mitochondrial damage and inflammatory cells exacerbate oxidative stress. Microdamage to the plasma membrane results in abnormal calcium influx, leading to mitochondrial dysfunction and activation of proteolytic enzymes that degrade muscle tissue. Endoplasmic reticulum activity is also disrupted by elevated sarcolipin levels. Dystrophin deficiency impairs satellite cell division and their ability to repair muscle damage, resulting in defective muscle regeneration and replacement of myocytes with fat and fibrotic tissue. Autophagy is also inhibited, leading to the accumulation of damaged organelles and proteins, and further muscle degeneration. Early in the disease, M1 macrophages and T lymphocytes infiltrate muscle tissue, exacerbating damage; at later stages, M2 macrophages predominate, promoting regeneration and fibrosis (Duan et al., 2021). In cardiac muscle, dystrophin interacts with specific protective proteins, which may explain the delayed onset of cardiomyopathy in DMD patients (Klietsch et al., 1993). Dystrophin is also expressed in specific brain structures, including the amygdala, hippocampus, and Purkinje cells. Its deficiency is associated with cognitive impairment and behavioral disturbances in approximately one-third of individuals with DMD (Thangarajh et al., 2019).

4.2 Current Standard Treatments and Their Limitations

At diagnosis, boys with Duchenne muscular dystrophy typically show substantial muscle mass loss, and available treatments - primarily physiotherapy and glucocorticoids-can only slow disease progression (Birnkrant et al., 2018). Glucocorticoids suppress the immune system and help prolong ambulation. Deflazacort was approved by the U.S. Food and Drug Administration (FDA) for DMD treatment in 2017. Clinical studies have shown that daily deflazacort use is associated with improved functional outcomes, particularly in older patients and in advanced disease stages (McDonald et al., 2023). Although only deflazacort has received FDA approval, prednisone and prednisolone are also widely used (Zelikovich et al., 2022). Other anti-inflammatory agents have been tested in clinical trials. One such drug is vamorolone, approved in 2023 by both the FDA and the European Medicines Agency (EMA). It effectively inhibits the pro-inflammatory NF-κB pathway without adversely affecting bone remodeling biomarkers, thereby not increasing the risk of osteoporosis or fractures (Guglieri et al., 2022). Promising results have also been reported for givinostat, a histone deacetylase inhibitor that regulates gene expression in muscle and satellite cells by modulating dysregulated nitric oxide signaling. Givinostat received FDA approval in March 2024 and, in April 2025, EMA issued a conditional recommendation for its use alongside corticosteroids in ambulatory DMD patients aged six years and older in the European Union (EU) (EMA, 2025b; Syed et al., 2025).

Since nonsense mutations account for approximately 10–15% of DMD cases, ataluren has been used to treat this subgroup (Bladen et al., 2015). This compound enables ribosomes to "read through" premature stop codons, facilitating the production of a truncated but partially functional dystrophin protein (Mercuri et al., 2023). Clinical trial results indicated modest improvements in treated boys, leading to conditional EMA approval in 2014 and limited use in non-EU countries-excluding the United States, where it was never approved. Although some newer data suggested a potential delay in symptom progression, EMA concluded that the drug's impact on dystrophin production was minimal. Consequently, the agency recommended against renewing marketing authorization in both 2023 and January 2024. Finally, on March 28, 2025, the European Commission revoked ataluren's conditional marketing authorization in the EU (Bello et al., 2025; EMA, 2025a).

Antisense oligonucleotides (ASOs) are synthetic nucleic acid fragments that modulate pre-mRNA splicing, promoting exon skipping in the dystrophin gene to produce a shortened yet functional dystrophin protein (Aoki & Wood, 2021). Four such drugs-eteplirsen, golodirsen, casimersen, and viltolarsen-received conditional approval in the U.S. between 2016 and 2021. Viltolarsen was also approved in Japan in 2020. None have been approved by the EMA. Although these therapies apply to only about 14% of DMD patients, they may improve muscle function despite modest increases in dystrophin levels (Roshmi & Yokota, 2021).

CRISPR/Cas9 technologies are currently under investigation as potential DMD treatments (Szwec et al., 2024). These methods enable dystrophin gene correction in animal models, restoring functional protein expression. Other experimental approaches aim to upregulate utrophin expression as a partial dystrophin substitute, although efficacy remains limited (Morin et al., 2023). This strategy only affects a subset of muscle nuclei, resulting in uneven tissue protection. Risks associated with AAV vector-based gene delivery—such as immune responses and off-target mutations- persist, though rAAV vectors continue to serve as the foundation for several approved gene therapies (Bennett, 2023; Bönnemann et al., 2023; Morin et al., 2023). Restoration of dystrophin function can be achieved by administering recombinant adeno-associated viral vectors (rAAV) carrying sequences encoding micro-dystrophin or micro-utrophin (Morin et al., 2023). These sequences are

based on functional variants of dystrophin observed in patients with the milder Becker muscular dystrophy (BMD) phenotype (Ramos et al., 2019). Alternatively, utrophin levels can be increased using molecules that modulate UTRN gene regulators or stabilize the utrophin-glycoprotein complex (UGC). Preliminary preclinical studies in mouse and dog models have demonstrated a high therapeutic potential of micro-dystrophin and micro-utrophin (μ Dys/ μ Utr). However, micro-dystrophin elicits a stronger immune response, which has been associated in some clinical trials with serious adverse events, such as immune-mediated myositis. On the other hand, while micro-utrophin has a more favorable safety profile, it exhibits lower mechanical efficacy and limited durability of expression in myocytes.

The first gene therapy to receive initial conditional approval from the FDA in 2023 for patients aged 4– 5 years was delandistrogene moxeparvovec (Zaidman et al., 2024). Delandistrogene moxeparvovec (Elevidys) is a gene therapy developed as a treatment for DMD. It comprises a non-replicating recombinant AAV vector derived from serotype 74 isolated from rhesus macaques (rAAVrh74), carrying a micro-dystrophin transgene, leading to the production of a truncated micro-dystrophin protein (with a molecular weight of 138 kDa, compared to 427 kDa for full-length dystrophin in healthy muscle) (Mendell, Sahenk, et al., 2024). Gene replacement therapies under development for Duchenne muscular dystrophy (DMD) utilize truncated constructs of the DMD gene ("micro-dystrophins") containing only the most essential domains, as the wildtype DMD gene, consisting of 79 exons and encoding a 14 kb transcript, exceeds the packaging capacity of adeno-associated virus (AAV) vectors (approximately 5 kb) [38]. As previously mentioned, the delandistrogene moxeparvovec gene therapy received accelerated approval from the U.S. Food and Drug Administration on June 22, 2023, for ambulatory boys aged 4–5 years. Subsequently, on June 20, 2024, based on phase 3 clinical trial data, the indication was expanded, resulting in full approval for use in ambulatory boys with DMD aged 4 years and older, along with accelerated approval for use in non-ambulatory patients with DMD (Bhattacharyya et al., 2024; Zaidman et al., 2024).

4.3 Effect of Elevidys Gene Therapy on Skeletal Muscle Function and Physical Performance in Patients with DMD

To date, results from three completed clinical trials and the first completed cohort (cohort 1) of the ongoing ENDEAVOR phase 1b study investigating delandistrogene moxeparvovec have been published. Their primary aim has been to assess the therapy's impact on motor function and physical performance in boys with DMD, particularly within the 4–8 year age range. Two additional studies are ongoing. Across all trials, the primary measure of therapeutic efficacy was the North Star Ambulatory Assessment (NSAA), which includes 17 tasks evaluating gait and gross motor function. Each task is scored on a three-point scale: 0 indicates inability to perform the task, 1 indicates performance with difficulty or assistance, and 2 indicates independent completion. The scale assesses activities such as rising from the floor, walking, and running. Secondary outcomes were also analyzed, including time to rise from the floor (TTR) and the 10-meter walk/run test (10MWR). In healthy children, the maximum score of 34 is typically achieved by age 4. In patients with DMD, improvement is observed until approximately age 6, followed by a gradual annual decline averaging 3.7 points after age 7. The minimally clinically important difference (MCID) on the NSAA is estimated to range between 2.3 and 3.5 points, depending on the calculation method (Ayyar Gupta et al., 2023).

The first study (NCT03375164) was a open-label, phase 1/2a, nonrandomized controlled trial involving four boys aged 4 to under 8 years with confirmed dystrophin gene mutations. The trial assessed the safety and tolerability of a single intravenous dose of delandistrogene moxeparvovec. Treatment was preceded by a dose of prednisone administered the day before infusion and continued for at least 30 days. After one year of follow-up, improvement in motor function was observed; however, the lack of a control group classifies this study as providing insufficient evidence of efficacy (Class IV). A subsequent publication included a comparison with an external control group matched via propensity score weighting, estimating a mean NSAA difference of 9.4 points in favor of gene therapy after four years (95% CI: 2.02–16.78), though this still constitutes Class III evidence (Mendell, Sahenk, et al., 2024).

The second clinical trial (NCT03769116) was a randomized, double-blind, placebo-controlled, phase 2 study conducted at two centers, divided into two 48-week parts. A total of 41 boys aged 4 to under 8 years with mutations involving exons 18–58 were enrolled. Participants were randomly assigned to receive either the gene therapy or placebo, with crossover in the second part of the study. All patients received prednisone for at least 60 days prior to therapy. The primary endpoint was the level of micro-dystrophin expression at week 12 post-treatment. This level, measured by Western blot, reached 23.82% of the physiological level in healthy individuals, indicating successful protein production. However, no statistically significant

improvement in NSAA scores was observed after 48 weeks across the full cohort-the mean difference between treated and control groups was 0.8 points (95% CI: -0.95 to 2.55), suggesting the result could be due to chance. Subgroup analysis by age revealed a significant improvement of 2.5 points (95% CI: 0.44-4.56) in children aged 4-5, while older children (6-7 years) showed a non-significant and slightly negative change (-0.7 points). After 96 weeks, the mean NSAA difference was 2.0 points (95% CI: -0.50 to 4.50), still not definitively significant. A similar result was obtained in the second part of the study when comparing with an external cohort-the LSM difference was 2.0 points (95% CI: 0.82-3.18), which this time reached statistical significance. randomized double-blind provides As а and study, this trial Class I evidence (Mendell et al., 2023).

The ENDEAVOR study is a two-part, open-label, phase 1b study of delandistrogene moxeparvovec. As of today, data from cohorts 2–7 of this study have not yet been published Cohort 1 of ENDEAVOR study comprising 20 ambulatory boys with DMD aged 4 to 8 years received a single intravenous dose of the therapy and corticosteroids for at least 60 days post-treatment. After one year, motor function was compared to a propensity score–matched control group (91 children). Results showed that treated boys improved by a mean of 3.2 NSAA points (95% CI: 1.59–4.81). Additionally, they were 1.2 seconds faster in the time to rise test (95% CI: -1.81 to -0.60) and 1 second faster in the 10-meter walk/run test (95% CI: -1.63 to -0.37). In a separate group of older, non-ambulatory boys (6 patients, mean age 15.26 years), upper limb function declined by -1.5 points after 52 weeks and -3.8 points after 104 weeks—indicating a slower decline than the natural disease course, which showed a -6.3 point decrease. Despite promising results, the lack of randomization categorizes this study as providing Class III, or moderate-quality, evidence(Zaidman et al., 2023).

The EMBARK trial (NCT05096221) is a phase 3, two-part, multinational, randomized, double-blind, placebo-controlled study involving 125 ambulatory DMD patients aged 4 to <8 years, with mutations in exons 18–44 and 46–79. Randomization was stratified by age and baseline NSAA score. Corticosteroid doses were comparable at baseline, but from weeks 2 to 12, the treatment group received higher doses. The primary endpoint—change in total NSAA score after 52 weeks-was not met (LSM difference 0.65; 95% CI: –0.45 to 1.75). No significant differences were observed across age or baseline NSAA subgroups. Secondary analyses showed small but significant improvements in functional tests: time to rise (TTR) and 10-meter walk (LSM differences -0.64 and -0.42 seconds, respectively). Step velocity and 4-stair climb time also favored the treatment group. The 100-meter walk/run test showed no difference. Patient-reported outcomes related to mobility and upper limb function were similar between groups. Nausea and vomiting occurred in 54% of treated participants and 0% in the placebo group, which may have compromised blinding integrity (Mendell et al., 2025). A comparison of the studies described is presented in Table 1.

Study (ID)	Study Characteristics	Number of Patients	Age of Patients	NSAA Change	Class of Evidence
NCT03375164	Phase 1/2a, open- label, nonrandomized controlled trial	4 boys	4–<8 years	+9.4 points (at 4 years vs. external group)	Class III evidence
NCT03769116	Phase 2, randomized, double-blind, placebo- controlled	41 boys	4–<8 years	+0.8 pts at 48 weeks (entire group, NS), +2.5 pts (age 4–5, IS), -0.7 pts (age 6–7, NS), +2.0 pts at 96 weeks (NS) vs. external group (IS)	Class I evidence
ENDEAVOR (NCT04626674)	Phase 1b, open-label	20 boys (cohort 1)	4–8 years, ambulatory	+3.2 pts after 1 year (vs. external group, IS)	Class III evidence
EMBARK (NCT05096221)	Phase 3, randomized, double-blind, placebo- controlled	125 boys	4–<8 years	+0.65 pts after 52 wks (NS)	Class I evidence

Table 1. Clinical efficacy of Delandistrogene moxeparvovec gene therapy across selected trials, focusing on changes in motor function as measured by the North Star Ambulatory Assessment (NSAA).

Based on clinical trial data, thirteen adverse events requiring treatment were identified, including vomiting, myocarditis, acute liver injury, and immune-mediated myositis (Zaidman et al., 2024). To mitigate the risk of immune response, corticosteroids are administered to patients one day prior to infusion and continued for at least 60 days thereafter. The most frequently observed adverse events (in \geq 5% of patients) include nausea, vomiting, fever, thrombocytopenia, and transient elevations in liver enzymes, which typically resolve during steroid therapy. Less common but more serious complications include acute hepatitis and myocarditis. One death due to liver failure was reported in a patient treated outside of a clinical trial setting. In some patients with deletions involving exons 8 or 9, myositis occurred, leading to marked weakness and difficulty swallowing or breathing. Safety data for individuals with deletions spanning exons 1–17 are lacking, as this group was not included in key clinical studies (Mendell, Proud, et al., 2024).

Delandistrogene moxeparvovec received accelerated approval from the FDA on June 22, 2023, for ambulatory boys aged 4–5 years (Zaidman et al., 2024). Clinical, pharmacological, and statistical review teams considered that the data did not meet the criteria for accelerated approval, as micro-dystrophin expression was not deemed a sufficient surrogate endpoint; consequently, the decision was considered controversial. Subsequently, on June 20, 2024, the FDA granted full approval for ambulatory patients aged \geq 4 years and accelerated approval for non-ambulatory patients aged \geq 4 years. This decision was made despite negative recommendations from several review divisions (including clinical and biostatistical offices). It was concluded that data from the EMBARK (NCT05096221) and ENDEAVOR (NCT04626674) trials provided reliable evidence of efficacy, with the exception of patients with exon 8 and/or 9 deletions, in whom the use of the drug is contraindicated. The increase in micro-dystrophin levels in non-ambulatory patients was considered potentially predictive of clinical benefit, forming the basis for approval despite the absence of proven clinical efficacy in this subgroup. Ongoing trials aim to provide further data, including in boys with other mutations and in non-ambulatory patients. Many of these individuals accept an elevated risk due to the lack of alternative disease-modifying therapies (Crossnohere et al., 2022; Oskoui et al., 2025).

5. CONCLUSIONS

Duchenne muscular dystrophy (DMD) is a severe, progressive neuromuscular disorder of genetic origin, caused by mutations in the dystrophin gene. The absence of functional dystrophin protein leads to muscle fiber degeneration, chronic inflammation, and gradual deterioration of muscle strength and function. Symptoms typically emerge in early childhood, and despite advances in supportive care, DMD still results in premature death and remains a major therapeutic challenge.

Delandistrogene moxeparvovec is an investigational gene therapy designed to deliver a shortened version of the dystrophin gene (micro-dystrophin) to muscle cells using an adeno-associated viral (AAV) vector. Although earlier phase studies suggested improved micro-dystrophin expression and potential clinical benefit, the phase 3 EMBARK trial results were less promising. The primary endpoint-change in motor function assessed by the North Star Ambulatory Assessment (NSAA) after 52 weeks-did not demonstrate statistically or clinically significant improvement compared to placebo.

The NSAA is one of the most commonly used functional scales in boys with DMD, evaluating the ability to perform daily activities that require strength and coordination. While useful, this scale may have limited sensitivity in detecting subtle, slow functional changes, particularly in studies with short observation periods. Furthermore, minor differences in secondary endpoints in the EMBARK trial also failed to reach statistical significance.

Interpretation of the results is complicated by the use of corticosteroids—the standard of care in DMD which independently improve muscle function, delay disease progression, and may mask or alter the response to gene therapy. Variability in dosing, duration, and type of steroid used represents a significant confounding factor that must be considered when analyzing treatment effects.

Ongoing studies are expected to provide more precise data regarding the long-term efficacy and safety of delandistrogene moxeparvovec. It is important to exercise caution in interpreting the current findings and to continue research efforts aimed at defining the true role of gene therapy in the treatment of DMD.

DISCLOSURE

Authors' contribution:

Conceptualization: Katarzyna Kozon, Agnieszka Floriańczyk Methodology: Agnieszka Ozdarska, Adrian Krzysztof Biernat Software: Anna Rupińska Check: Adrian Krzysztof Biernat Formal analysis: Agnieszka Floriańczyk, Ewa Romanowicz Investigation: Marcin Lampart, Zuzanna Jasińska Resources: Aleksandra Kołdyj, Kamila Krzewska Data curation: Agnieszka Ozdarska Writing -rough preparation: Katarzyna Kozon, Zuzanna Jasińska Writing -review and editing: Ewa Romanowicz, Kamila Krzewska Supervision: Katarzyna Kozon Project administration: Aleksandra Kołdyj, Marcin Lampart All authors have read and agreed with the published version of the manuscript.

Receiving funding: not applicable;
Funding Statement: The study did not receive special funding.
Institutional Review Board Statement: Not applicable.
Informed Consent Statement: Not applicable.
Data Availability Statement: Not applicable.
Conflicts of Interest: The authors declare no conflicts of interest.
Acknowledgements: Not applicable.

REFERENCES

- 1. Aartsma-Rus, A., & van Putten, M. (2014). Assessing functional performance in the mdx mouse model. J Vis Exp(85). https://doi.org/10.3791/51303
- 2. Aldharee, H. (2024). Duchenne muscular dystrophy in Saudi Arabia: a review of the current literature. *Front Neurol*, *15*, 1392274. https://doi.org/10.3389/fneur.2024.1392274
- Aoki, Y., & Wood, M. J. A. (2021). Emerging Oligonucleotide Therapeutics for Rare Neuromuscular Diseases. J Neuromuscul Dis, 8(6), 869-884. https://doi.org/10.3233/jnd-200560
- Ayyar Gupta, V., Pitchforth, J. M., Domingos, J., Ridout, D., Iodice, M., Rye, C., Chesshyre, M., Wolfe, A., Selby, V., Mayhew, A., Mazzone, E. S., Ricotti, V., Hogrel, J. Y., Niks, E. H., de Groot, I., Servais, L., Straub, V., Mercuri, E., Manzur, A. Y., & Muntoni, F. (2023). Determining minimal clinically important differences in the North Star Ambulatory Assessment (NSAA) for patients with Duchenne muscular dystrophy. *PLoS One*, *18*(4), e0283669. https://doi.org/10.1371/journal.pone.0283669
- Bello, L., Riguzzi, P., Albamonte, E., Astrea, G., Battini, R., Barp, A., Berardinelli, A. L., Bertini, E. S., Brolatti, N., Bruno, C., Corti, S., D'Amico, A., D'Angelo, M. G., Dallavalle, G., Liguori, R., Maggi, L., Magri, F., Mancuso, M., Masson, R., Mercuri, E., Minetti, C., Messina, S., Mongini, T., Musumeci, O., Nigro, V., Pane, M., Panicucci, C., Pedemonte, M., Pini, A., Politano, L., Previtali, S., Ricci, F., Ricci, G., Ruggiero, L., Sansone, V., Siciliano, G., Trabacca, A., Trucco, F., Velardo, D., Pegoraro, E., & Comi, G. P. (2025). Opinion of the Italian Association of Myology on Ataluren for the Treatment of Nonsense Mutation Duchenne Muscular Dystrophy. *Drugs R D*, 25(2), 99-106. https://doi.org/10.1007/s40268-025-00512-x
- 6. Bennett, J. (2023). Overview of Retinal Gene Therapy: Current Status and Future Challenges. *Cold Spring Harb Perspect Med*, *13*(7). https://doi.org/10.1101/cshperspect.a041278
- Bez Batti Angulski, A., Hosny, N., Cohen, H., Martin, A. A., Hahn, D., Bauer, J., & Metzger, J. M. (2023). Duchenne muscular dystrophy: disease mechanism and therapeutic strategies. *Front Physiol*, 14, 1183101. https://doi.org/10.3389/fphys.2023.1183101
- 8. Bhattacharyya, M., Miller, L. E., Miller, A. L., & Bhattacharyya, R. (2024). The FDA approval of delandistrogene moxeparvovec-rokl for Duchenne muscular dystrophy: a critical examination of the evidence and regulatory process. *Expert Opin Biol Ther*, *24*(9), 869-871. https://doi.org/10.1080/14712598.2024.2394477
- Birnkrant, D. J., Bushby, K., Bann, C. M., Apkon, S. D., Blackwell, A., Brumbaugh, D., Case, L. E., Clemens, P. R., Hadjiyannakis, S., Pandya, S., Street, N., Tomezsko, J., Wagner, K. R., Ward, L. M., & Weber, D. R. (2018). Diagnosis and management of Duchenne muscular dystrophy, part 1: diagnosis, and neuromuscular, rehabilitation, endocrine, and gastrointestinal and nutritional management. *Lancet Neurol*, *17*(3), 251-267. https://doi.org/10.1016/s1474-4422(18)30024-3

- Bladen, C. L., Salgado, D., Monges, S., Foncuberta, M. E., Kekou, K., Kosma, K., Dawkins, H., Lamont, L., Roy, A. J., Chamova, T., Guergueltcheva, V., Chan, S., Korngut, L., Campbell, C., Dai, Y., Wang, J., Barišić, N., Brabec, P., Lahdetie, J., Walter, M. C., Schreiber-Katz, O., Karcagi, V., Garami, M., Viswanathan, V., Bayat, F., Buccella, F., Kimura, E., Koeks, Z., van den Bergen, J. C., Rodrigues, M., Roxburgh, R., Lusakowska, A., Kostera-Pruszczyk, A., Zimowski, J., Santos, R., Neagu, E., Artemieva, S., Rasic, V. M., Vojinovic, D., Posada, M., Bloetzer, C., Jeannet, P. Y., Joncourt, F., Díaz-Manera, J., Gallardo, E., Karaduman, A. A., Topaloğlu, H., El Sherif, R., Stringer, A., Shatillo, A. V., Martin, A. S., Peay, H. L., Bellgard, M. I., Kirschner, J., Flanigan, K. M., Straub, V., Bushby, K., Verschuuren, J., Aartsma-Rus, A., Béroud, C., & Lochmüller, H. (2015). The TREAT-NMD DMD Global Database: analysis of more than 7,000 Duchenne muscular dystrophy mutations. *Hum Mutat*, *36*(4), 395-402. https://doi.org/10.1002/humu.22758
- 11. Bönnemann, C. G., Belluscio, B. A., Braun, S., Morris, C., Singh, T., & Muntoni, F. (2023). Dystrophin Immunity after Gene Therapy for Duchenne's Muscular Dystrophy. *N Engl J Med*, 388(24), 2294-2296. https://doi.org/10.1056/NEJMc2212912
- 12. Chelly, J., Hamard, G., Koulakoff, A., Kaplan, J. C., Kahn, A., & Berwald-Netter, Y. (1990). Dystrophin gene transcribed from different promoters in neuronal and glial cells. *Nature*, *344*(6261), 64-65. https://doi.org/10.1038/344064a0
- Crossnohere, N. L., Fischer, R., Vroom, E., Furlong, P., & Bridges, J. F. P. (2022). A Comparison of Caregiver and Patient Preferences for Treating Duchenne Muscular Dystrophy. *Patient*, 15(5), 577-588. https://doi.org/10.1007/s40271-022-00574-y
- 14. Duan, D., Goemans, N., Takeda, S., Mercuri, E., & Aartsma-Rus, A. (2021). Duchenne muscular dystrophy. *Nat Rev Dis Primers*, 7(1), 13. https://doi.org/10.1038/s41572-021-00248-3
- 15. EMA. (2025a). *EMA decision to not renew the conditional marketing authorisation for Translarna (ataluren) in the European* Union (EU). Retrieved 09.07 from https://www.ema.europa.eu/en/medicines/human/EPAR/translarna?utm source
- 16. EMA. (2025b). *New treatment against Duchenne muscular dystrophy* https://www.ema.europa.eu/en/news/new-treatment-against-duchenne-muscular-dystrophy?utm source
- 17. Gao, Q. Q., & McNally, E. M. (2015). The Dystrophin Complex: Structure, Function, and Implications for Therapy. *Compr Physiol*, 5(3), 1223-1239. https://doi.org/10.1002/cphy.c140048
- Górecki, D. C., Monaco, A. P., Derry, J. M., Walker, A. P., Barnard, E. A., & Barnard, P. J. (1992). Expression of four alternative dystrophin transcripts in brain regions regulated by different promoters. *Hum Mol Genet*, 1(7), 505-510. https://doi.org/10.1093/hmg/1.7.505
- Guglieri, M., Clemens, P. R., Perlman, S. J., Smith, E. C., Horrocks, I., Finkel, R. S., Mah, J. K., Deconinck, N., Goemans, N., Haberlova, J., Straub, V., Mengle-Gaw, L. J., Schwartz, B. D., Harper, A. D., Shieh, P. B., De Waele, L., Castro, D., Yang, M. L., Ryan, M. M., McDonald, C. M., Tulinius, M., Webster, R., McMillan, H. J., Kuntz, N. L., Rao, V. K., Baranello, G., Spinty, S., Childs, A. M., Sbrocchi, A. M., Selby, K. A., Monduy, M., Nevo, Y., Vilchez-Padilla, J. J., Nascimento-Osorio, A., Niks, E. H., de Groot, I. J. M., Katsalouli, M., James, M. K., van den Anker, J., Damsker, J. M., Ahmet, A., Ward, L. M., Jaros, M., Shale, P., Dang, U. J., & Hoffman, E. P. (2022). Efficacy and Safety of Vamorolone vs Placebo and Prednisone Among Boys With Duchenne Muscular Dystrophy: A Randomized Clinical Trial. *JAMA Neurol*, *79*(10), 1005-1014. https://doi.org/10.1001/jamaneurol.2022.2480
- Klietsch, R., Ervasti, J. M., Arnold, W., Campbell, K. P., & Jorgensen, A. O. (1993). Dystrophin-glycoprotein complex and laminin colocalize to the sarcolemma and transverse tubules of cardiac muscle. *Circ Res*, 72(2), 349-360. https://doi.org/10.1161/01.res.72.2.349
- McDonald, C. M., Marden, J. R., Shieh, P. B., Wong, B. L., Lane, H., Zhang, A., Nguyen, H., Frean, M., Trifillis, P., Koladicz, K., & Signorovitch, J. (2023). Disease progression rates in ambulatory Duchenne muscular dystrophy by steroid type, patient age and functional status. *J Comp Eff Res*, 12(4), e220190. https://doi.org/10.57264/cer-2022-0190
- Mendell, J. R., Muntoni, F., McDonald, C. M., Mercuri, E. M., Ciafaloni, E., Komaki, H., Leon-Astudillo, C., Nascimento, A., Proud, C., Schara-Schmidt, U., Veerapandiyan, A., Zaidman, C. M., Guridi, M., Murphy, A. P., Reid, C., Wandel, C., Asher, D. R., Darton, E., Mason, S., Potter, R. A., Singh, T., Zhang, W., Fontoura, P., Elkins, J. S., & Rodino-Klapac, L. R. (2025). AAV gene therapy for Duchenne muscular dystrophy: the EMBARK phase 3 randomized trial. *Nat Med*, *31*(1), 332-341. https://doi.org/10.1038/s41591-024-03304-z
- Mendell, J. R., Proud, C., Zaidman, C. M., Mason, S., Darton, E., Wang, S., Wandel, C., Murphy, A. P., Mercuri, E., Muntoni, F., & McDonald, C. M. (2024). Practical Considerations for Delandistrogene Moxeparvovec Gene Therapy in Patients With Duchenne Muscular Dystrophy. *Pediatr Neurol*, 153, 11-18. https://doi.org/10.1016/j.pediatrneurol.2024.01.003
- Mendell, J. R., Sahenk, Z., Lehman, K. J., Lowes, L. P., Reash, N. F., Iammarino, M. A., Alfano, L. N., Lewis, S., Church, K., Shell, R., Potter, R. A., Griffin, D. A., Hogan, M., Wang, S., Mason, S., Darton, E., & Rodino-Klapac, L. R. (2024). Long-term safety and functional outcomes of delandistrogene moxeparvovec gene therapy in patients with Duchenne muscular dystrophy: A phase 1/2a nonrandomized trial. *Muscle Nerve*, 69(1), 93-98. https://doi.org/10.1002/mus.27955

- 25. Mendell, J. R., Shieh, P. B., McDonald, C. M., Sahenk, Z., Lehman, K. J., Lowes, L. P., Reash, N. F., Iammarino, M. A., Alfano, L. N., Sabo, B., Woods, J. D., Skura, C. L., Mao, H. C., Staudt, L. A., Griffin, D. A., Lewis, S., Wang, S., Potter, R. A., Singh, T., & Rodino-Klapac, L. R. (2023). Expression of SRP-9001 dystrophin and stabilization of motor function up to 2 years post-treatment with delandistrogene moxeparvovec gene therapy in individuals with Duchenne muscular dystrophy. Front Cell Dev Biol. 11, 1167762. https://doi.org/10.3389/fcell.2023.1167762
- 26. Mercuri, E., Bönnemann, C. G., & Muntoni, F. (2019). Muscular dystrophies. *Lancet*, 394(10213), 2025-2038. https://doi.org/10.1016/s0140-6736(19)32910-1
- Mercuri, E., Osorio, A. N., Muntoni, F., Buccella, F., Desguerre, I., Kirschner, J., Tulinius, M., de Resende, M. B. D., Morgenroth, L. P., Gordish-Dressman, H., Johnson, S., Kristensen, A., Werner, C., Trifillis, P., Henricson, E. K., & McDonald, C. M. (2023). Safety and effectiveness of ataluren in patients with nonsense mutation DMD in the STRIDE Registry compared with the CINRG Duchenne Natural History Study (2015-2022): 2022 interim analysis. *J Neurol*, 270(8), 3896-3913. https://doi.org/10.1007/s00415-023-11687-1
- Morin, A., Stantzou, A., Petrova, O. N., Hildyard, J., Tensorer, T., Matouk, M., Petkova, M. V., Richard, I., Manoliu, T., Goyenvalle, A., Falcone, S., Schuelke, M., Laplace-Builhé, C., Piercy, R. J., Garcia, L., & Amthor, H. (2023). Dystrophin myonuclear domain restoration governs treatment efficacy in dystrophic muscle. *Proc Natl Acad Sci U* S A, 120(2), e2206324120. https://doi.org/10.1073/pnas.2206324120
- 29. Nevin, N. C., Hughes, A. E., Calwell, M., & Lim, J. H. (1986). Duchenne muscular dystrophy in a female with a translocation involving Xp21. *J Med Genet*, 23(2), 171-173. https://doi.org/10.1136/jmg.23.2.171
- Oskoui, M., Caller, T. A., Parsons, J. A., Servais, L., Butterfield, R. J., Bharadwaj, J., Rose, S. C., Tolchin, B., Puskala Hamel, K., Silsbee, H. M., & Dowling, J. J. (2025). Delandistrogene Moxeparvovec Gene Therapy in Individuals With Duchenne Muscular Dystrophy: Evidence in Focus. *Neurology*, 104(11), e213604. https://doi.org/doi:10.1212/WNL.000000000213604
- 31. Patterson, G., Conner, H., Groneman, M., Blavo, C., & Parmar, M. S. (2023). Duchenne muscular dystrophy: Current treatment and emerging exon skipping and gene therapy approach. *Eur J Pharmacol*, 947, 175675. https://doi.org/10.1016/j.ejphar.2023.175675
- 32. Ramos, J. N., Hollinger, K., Bengtsson, N. E., Allen, J. M., Hauschka, S. D., & Chamberlain, J. S. (2019). Development of Novel Micro-dystrophins with Enhanced Functionality. *Mol Ther*, 27(3), 623-635. https://doi.org/10.1016/j.ymthe.2019.01.002
- Roshmi, R. R., & Yokota, T. (2021). Pharmacological Profile of Viltolarsen for the Treatment of Duchenne Muscular Dystrophy: A Japanese Experience. *Clin Pharmacol*, 13, 235-242. https://doi.org/10.2147/cpaa.S288842
- Sander, M., Chavoshan, B., Harris, S. A., Iannaccone, S. T., Stull, J. T., Thomas, G. D., & Victor, R. G. (2000). Functional muscle ischemia in neuronal nitric oxide synthase-deficient skeletal muscle of children with Duchenne muscular dystrophy. *Proc Natl Acad Sci U S A*, 97(25), 13818-13823. https://doi.org/10.1073/pnas.250379497
- 35. Shih, J. A., Folch, A., & Wong, B. L. (2020). Duchenne Muscular Dystrophy: the Heart of the Matter. *Curr Heart Fail Rep*, *17*(3), 57-66. https://doi.org/10.1007/s11897-020-00456-0
- Syed, N., Un Nisa Mughal, Z., Haseeb, A., Alemam Manhal, G., Yasir Eissa, A., & Abbasher Hussien Mohamed Ahmed, K. (2025). Duvyzat (givinostat): a new hope for Duchenne muscular dystrophy patients. *Ann Med Surg* (Lond), 87(5), 2529-2531. https://doi.org/10.1097/ms9.00000000003196
- Szwec, S., Kapłucha, Z., Chamberlain, J. S., & Konieczny, P. (2024). Dystrophin- and Utrophin-Based Therapeutic Approaches for Treatment of Duchenne Muscular Dystrophy: A Comparative Review. *BioDrugs*, 38(1), 95-119. https://doi.org/10.1007/s40259-023-00632-3
- Thangarajh, M., Hendriksen, J., McDermott, M. P., Martens, W., Hart, K. A., & Griggs, R. C. (2019). Relationships between DMD mutations and neurodevelopment in dystrophinopathy. *Neurology*, 93(17), e1597-e1604. https://doi.org/10.1212/wnl.00000000008363
- Trucco, F., Domingos, J. P., Tay, C. G., Ridout, D., Maresh, K., Munot, P., Sarkozy, A., Robb, S., Quinlivan, R., Riley, M., Burch, M., Fenton, M., Wallis, C., Chan, E., Abel, F., Manzur, A. Y., & Muntoni, F. (2020). Cardiorespiratory Progression Over 5 Years and Role of Corticosteroids in Duchenne Muscular Dystrophy: A Single-Site Retrospective Longitudinal Study. *Chest*, 158(4), 1606-1616. https://doi.org/10.1016/j.chest.2020.04.043
- Zaidman, C. M., Goedeker, N. L., Aqul, A. A., Butterfield, R. J., Connolly, A. M., Crystal, R. G., Godwin, K. E., Hor, K. N., Mathews, K. D., Proud, C. M., Kula Smyth, E., Veerapandiyan, A., Watkins, P. B., & Mendell, J. R. (2024). Management of Select Adverse Events Following Delandistrogene Moxeparvovec Gene Therapy for Patients With Duchenne Muscular Dystrophy. *J Neuromuscul Dis*, 11(3), 687-699. https://doi.org/10.3233/jnd-230185
- Zaidman, C. M., Proud, C. M., McDonald, C. M., Lehman, K. J., Goedeker, N. L., Mason, S., Murphy, A. P., Guridi, M., Wang, S., Reid, C., Darton, E., Wandel, C., Lewis, S., Malhotra, J., Griffin, D. A., Potter, R. A., Rodino-Klapac, L. R., & Mendell, J. R. (2023). Delandistrogene Moxeparvovec Gene Therapy in Ambulatory Patients (Aged ≥4 to <8 Years) with Duchenne Muscular Dystrophy: 1-Year Interim Results from Study SRP-9001-103 (ENDEAVOR). *Ann Neurol*, 94(5), 955-968. https://doi.org/10.1002/ana.26755
- Zelikovich, A. S., Joslin, B. C., Casey, P., McNally, E. M., & Ajroud-Driss, S. (2022). An Open Label Exploratory Clinical Trial Evaluating Safety and Tolerability of Once-Weekly Prednisone in Becker and Limb-Girdle Muscular Dystrophy. J Neuromuscul Dis, 9(2), 275-287. https://doi.org/10.3233/jnd-210741